Author’s response to reviews

Title: Profile and course of early Rheumatoid Arthritis in Morocco. A two-year follow-up study.

Authors:

benbouazza Karima (kbenbouazza@yahoo.fr)
benchekroun bahia (benchek_1@yahoo.fr)
rkain hanan (hananrkain@yahoo.fr)
amine bouchra (amine_bouchra@yahoo.fr)
bzami fatiha (bzamifatiha@yahoo.fr)
benbrahim leila (l.benbrahim@yahoo.fr)
atouf ouafa (atoufoufa@gmail.com)
essakalli malika (m_essakalli@hotmail.com)
abouqal redouane (abouqal@invivo.edu)
dougados maxime (m.doug@cch.aphp.fr)
hajjaj-hassouni najia (nhajjajhassouni@gmail.com)

Version: 2 Date: 21 July 2011

Author’s response to reviews:

AS RECOMMENDED BY THE EDITOR AND REVIEWERS, WE ADRESS THE REVISED MANUSCRIPT WITH ALL CHANGES MADE WITH COLOURED TEXT.